Author's response to reviews

Title: Birth preparedness and complication readiness among child bearing age women in Goba woreda, Oromia region, Ethiopia

Authors:

Desalegn Markos (desalegnmarkos@yahoo.com)
Daniel Bogale (dbogale386@gmail.com)

Version: 8 Date: 31 July 2014

Author's response to reviews:

Point by point response for Editorial Comment:
"Copyedit:

We recommend that you copyedit the paper to improve the style of written English. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz. Use of an editing service is neither a requirement nor a guarantee of acceptance for publication.

Response: two colleagues who have scientific expertise and are excellent in English language were consulted and edited the paper. Furthermore, both authors made their best effort to further edit the paper to improve the style of written English.

Please also ensure that your revised manuscript conforms to the journal style. It is important that your files are correctly formatted.

Response: the comment is accepted and the journal style as well as the format is cross checked and corrected accordingly.